\documentclass{jarticle}
\usepackage{overcite}
\usepackage{cite}
\usepackage{url}
\usepackage[dvipdfm,bookmarks=true,bookmarksnumbered=true,%
pdftitle={title},%
pdfsubject={subject},%
pdfauthor={美咲},%
pdfkeywords={keyword}
]{hyperref}
%\AtBeginDvi{\special{pdf:tounicode 90ms-RKSJ-UCS2}} % SJIS → Unicode
\usepackage{pxjahyper}


\renewcommand\citeform[1]{#1)}

\def\People#1#2{#1 #2\index{people\levelchar#2, #1\string|usage}}
\newcommand*\Color[2][+]{\textsl{#2}\ifx#1+\index{color names\levelchar\textsl{#2}\string|usage}\fi}
\def\Package#1{\textsf{#1}\index{packages\levelchar\textsf{#1}\string|usage}}
\def\CTAN#1#2{\texttt{\href{http://www.ctan.org/tex-archive/#1}{CTAN/#1#2}}}
\def\URL#1{\texttt{\url{#1}}}

\begin{document}
\begin{thebibliography}{9}

\bibitem{xcolor}%
 {\People{Uwe}{Kern}: \Package{xcolor} package, ``\LaTeX{} color extensions''.
  \CTAN{macros/latex/contrib/xcolor/}{} and
  \URL{http://www.ukern.de/tex/xcolor.html}}

\bibitem{lls80}
Lee, S., Locke, C. and Spurrier, J. (1980).
"On a class of tests of exponentiality"
{\sl Technometrics} \textbf{22} 547--554.
 
\bibitem{lzj}
Li, X., Zhong, C. and Jing, B. (2000). 
"Test for new better than used in convex ordering"
{\sl Communications in Statistics-Theory and Methods}, \textbf{29}(12), 2751--2760.
 


\bibitem{kum87}
Y.~KUMAZAWA and Y.~KUMAZAWA. A note on an estimator of life expectancy with random censorship. 
{\sl Biometrika.} \textbf{47}, 655-658. 1987. 

\bibitem{kum88}
Y.~KUMAZAWA and Y.~KUMAZAWA. A note on an estimator of life expectancy with random censorship. 
{\sl Biometrika.} \textbf{47}, 655-658. 1987. 

\end{thebibliography}


\section{A}


色は匂へど散りぬるを\cite{lls80,lzj,kum87,kum88}
我が世誰ぞつねならむ\cite{xcolor}





\end{document}

